Happiness and Life Satisfaction in Turkey in Recent Years

Asena Caner 1,*

Phone 90-312 292 4111

Email acaner@etu.edu.tr

¹ Department of Economics, TOBB University of Economics and Technology, Sogutozu Cad. No. 43, Sogutozu, 06560 Ankara, Turkey

Abstract

This paper aims to investigate the determinants of happiness and life satisfaction in Turkey in recent years. It uses regression analyses based on micro data from two nationally representative household surveys, namely the Turkish Life Satisfaction Survey and the World Values Survey for Turkey, in years 2007 and 2011. The wide variety of data collected by the two surveys enables a comprehensive analysis of the correlates of the outcome variable. Although some of the results are consistent both with the earlier ones in the literature and across the datasets used in this study, some others depend on the time period, the set of control variables and the sample used; thus casting doubt on the strength of the findings. To make a proper comparison and to isolate the differences generated by different samples, the analysis is restricted to the same set of control variables that are defined in the most similar manner. The variables whose estimates have been found to be similar are unemployment status, marital status, relative income, and gender. The differences are observed in the estimates of age, absolute income and education. The conclusion is that the question of what determines happiness is still a valid one that needs to be studied with more and better data. Recommendations are made for ways to improve data quality.

Keywords

Happiness
Life satisfaction
Determinants
Survey data
Turkey

1. Introduction

It is now widely recognized that traditional economic measures are insufficient to reflect a nation's well-being and that wider measures are needed. In that respect, the concepts of happiness and life satisfaction have come to the forefront. As policymakers seek to better understand both individual content and social stress, data have been collected in many countries on how people evaluate their own lives. Understanding and improving happiness and life satisfaction requires the collection of reliable evidence that can be used to inform policymakers. In fact, the body of evidence is growing fast, which can be judged easily by looking at the number of papers published in academic journals on the topic. Yet, it is not clearly known whether the identified determinants of happiness and life satisfaction can survive the test of time, or whether they are specific to the analyzed sample.

The purpose of this paper is to investigate the determinants of happiness and life satisfaction in Turkey in recent years by using the latest waves of nationally representative micro datasets. Two household surveys, the Turkish Life Satisfaction Survey (TLSS) and the World Values Survey for Turkey (WVS), in years 2007 and 2011, are used to conduct regression analyses. The wide variety of data collected by the two surveys enables a comprehensive analysis of the correlates of happiness and life satisfaction. The main questions asked are it is questioned whether the findings reported in earlier studies are still valid and whether the results obtained from the two different surveys are consistent with each other.

The paper contributes to the literature mainly in three ways: First, it analyzes comprehensively the determinants of happiness and life satisfaction in Turkey in recent years. Turkey is a developing country with a population of about 75

million. It is an OECD country in which average life satisfaction is unfortunately below the OECD average. Although Turkish economy has expanded considerably in the last decades, it appears that the people are not particularly satisfied with their lives. The country is similar in life satisfaction to Hungary, Portugal, Estonia, China, South Africa and Indonesia (OECD 2011), some of which have much lower per capita income than Turkey. Another ranking, in which the highest worldwide happiness score is 8.5 and the lowest is 2.6, assigns Turkey a score of 5.7 that places the country somewhere in the middle. In that scale, the EU average is 6.46 which is higher than the score of Turkey (Veenhoven 2013). This paper uses micro data from two large-scale nationally representative surveys. The surveys differ in the set of questions asked, thereby providing us with an opportunity to study an expanded set of variables, which is an advantage of the paper.

The second contribution of the paper is that it compares the estimates across the two surveys (the WVS and the TLSS samples) to see if the findings are robust to different samples of the Turkish population. To make a proper comparison, the regression equations are specified in the same manner with similarly defined control variables. The results tell us that although some estimates are similar, many estimates differ across the two surveys and across time. Such a result casts doubt on the findings reported in the literature.

Thirdly, the paper provides a list of recommendations for future work on the economics of happiness. The list focuses on ways to improve data quality as explained in detail in the paper. First, data collectors could be more sensitive to being consistent in the scale and the wording of the happiness questions across surveys. Secondly, the data could have a panel dimension. Thirdly, the sample size could be increased to achieve representativeness in regions and provinces. Turkey is a country with a large population and ample diversity across groups of people and regions. The variation across geographical areas could be employed to better understand the determinants of happiness and life satisfaction. It is inevitable that a renovation of the data collection process will increase costs, yet the increase in costs should be weighed against the substantial increase in benefits.

The plan of the paper is as follows. Section 2 introduces to the reader the relevant literature. Section 3 describes the datasets and the estimation

method. Section 4 explains the findings of the paper. Section 5 discusses the results and concludes.

2. Background

Happiness is a concept that all of us are familiar with, but may have a hard time defining. According to some authors, happiness is the main, if not the only, purpose of life (Ng 1996, p.1). The word 'happiness' is used in the widest sense as an umbrella term for all worth. In this meaning it is used interchangeably with terms like 'wellbeing' or 'quality of life', denoting both individual and social welfare. The word is also used more specifically to denote overall subjective evaluation of life as a whole (Veenhoven 2000) and defined concisely as "... how much one likes the life one lives".

Life satisfaction is conceptualized by some researchers as a subcategory of happiness (Davis and Fine-Davis 1991, p. 111). Some others see happiness and life satisfaction as two areas of psychological well-being (Zapf et al. 1987, p. 25). In the second approach, happiness is an emotional state produced by positive and negative events and experiences in life. Life satisfaction, on the other hand, is conceptualized as more of a cognitive evaluation that is somewhat dependent on social comparisons with reference groups as well as on the individual's desires, expectations, and hopes.

Typically, data on happiness and life satisfaction are harvested by means of survey questions such as: "Thinking about your life in general, how happy do you feel?" or "All things considered, how satisfied are you with your life as a whole in these days?" Both the definitions and the survey questions reflect subjectivity. The subjective approach to defining well-being runs counter to the objective approach of neoclassical economics, which relies on revealed preference and on observable choices of individuals to gather information required to measure well-being. The objective approach maintains that subjectivist experience gathered by survey questions is useless since it is not objectively observable (Frey and Stutzer 2002). Amartya Sen attributes the popularity of objectivity to "... a mixture of an obsessive concern with observability and a peculiar belief that choice ... is the only human aspect that can be observed." (Sen 1986, p. 18).

Advances in economics and psychology have shown us that choices are not the only human aspect that can be observed; the hedonic experience (such as pain, joy or pleasure) can also be observed. Experiments conducted by psychologists have revealed that decision utility and experienced utility are two separate concepts (Kahneman et al. 1997, p. 376). To determine an individual's experienced utility, we can simply ask him about his feelings since he is the best person to make a judgment. The subjective evaluation of the person can be used as a measure of utility (Frey and Stutzer 2002).

Since happiness and life satisfaction are the ultimate goals of life for many people, it is important to know what determines them, both for intrinsic interest and for economic policy. A new branch of economics, 'the economics of happiness', has emerged in the last two decades mainly as an empirical branch of economics. A large volume of research has already been conducted on large datasets where many factors are considered and controlled for. The entire literature will not be reviewed here since excellent reviews (and a defense on the validity of the approach) can be seen in Frey and Stutzer (2002), Clark et al. (2008), and Dolan et al. (2008). Most of the research has focused on exploring the determinants. The potential determinants studied in the literature are income (including absolute income, relative income and expected future income), personal characteristics (such as age, gender, ethnicity and personality), socially developed characteristics (such as education, health, unemployment, and the type of work done), work and daily activities (such as the number of hours worked, commuting, caring for others, community involvement and volunteering, exercise, and religious activities), attitudes and beliefs (such as trust, religion, and political view), relationships (such as marriage and intimacy, having children, seeing family and friends), and the factors related to the wider economic, social and political environment we live in (such as income inequality, unemployment rate, inflation, welfare system and public insurance, degree of democracy, climate and natural environment, safety, and urbanization).

The literature review made by Dolan et al. (2008) has revealed that the association of happiness with some factors, namely age, separation, unemployment and health, has been confirmed by using different data sets, different countries, different time periods, and different methods of analysis.

In fact, many of the findings reported in the literature on the determinants of happiness are what one expects to see. For example, significant differences in happiness are observed between those with different objective circumstances; such as employed versus unemployed, or single versus living with a partner. Nonetheless, the association of happiness with many other factors is ambiguous, varying across time, datasets, countries and methods of analysis. Comparing results across studies is difficult for two main reasons. First, a different set of control variables may be used and the estimated effect of a particular factor (both the coefficient size and its significance level) may depend on what other variables are controlled for in the regression equation. Secondly, categorical variables and reference categories may be defined differently across studies. For example, surveys may categorize educational status differently, treating the consecutive levels of education together or separately.

The literature on happiness and life satisfaction in Turkey contains only a few published studies. Gitmez and Morcol (1994) find that socio-economic status is a strong determinant of the vertical inequalities in the distribution of satisfaction in various domains of life (family, work, neighbors, etc.) using data from 145 face-to-face interviews in Ankara. Ekici and Koydemir (2014) investigate both the changes in happiness and life satisfaction in Turkey and how the two variables are related to social capital, using data from 1999 and 2008 waves of the European Values Study (EVS). They report significant increase in happiness, life satisfaction, and many social capital variables.

The studies that can be most easily related to this paper are Selim (2008), Dumludag (2013), and Caner (2014). Selim (2008) uses the 1990, 1996 and 2001 waves of WVS to study the determinants of happiness and life satisfaction. In this paper, I use the 2007 and 2011 waves (the most recent two waves) of the WVS and the same specification of the regression equation are used with the purpose of testing if the earlier findings are robust across time. Dumludag (2013) uses a different data source, the 2010 Life in Transition Survey, and focuses on how life satisfaction is influenced by income comparisons [both to others in the relevant reference group and to oneself in the past (evaluation) and future (expectation)]. It includes three comparison variables (own living standards in comparison to parents, current living standards in comparison to 4 years ago, and the position on the well-

being ladder) as control variables. In this paper, I use similarly defined comparison variables in the TLSS are used with the purpose of testing if the findings are robust across datasets. Both Selim (2008) and Dumludag (2013) use the ordered logit model to estimate happiness equations, but they differ in the set of control variables and variable definitions. In Sect. 4 of this paper, I show that the two studies yield some conflicting results; then, I explain which findings are robust across time and surveys. Caner (2014) uses data from TLSS in years 2003–2011 to ask whether income comparisons and expectations about future income and working conditions direct determine an individual's utility (proxied by happiness). The effects of comparisons and expectations on utility are related to the effect of income on utility. With such as focus, demographic variables are barely mentioned. The study relates to Caner (2014) this paper as it also includes comparison and expectation variables.

Since the purpose of this paper is to examine the determinants of happiness in Turkey in recent years, it is important to use, as has been done in this paper, to use data that are representative of the population. Another purpose of this paper is to know the extent to which the findings obtained from different samples and time periods are consistent. Dolan et al. (2008) report that "... the existing evidence base is not quite as strong as some people may have suggested..." (p. 112), rendering clear conclusions impossible and stressing the importance of conducting further analyses. As a contribution to the literature, this paper compares the findings from two separate datasets on happiness in Turkey, after making sure that the same set of variables are controlled for and that the variables are defined as similarly as possible.

3. Data and Methods

3.1. Data

The datasets used in this paper are the TLSS conducted by the Turkish Statistical Institute and the WVS conducted by the WVS Association, a non-profit research organization.

The first dataset, the TLSS, is a nationally representative survey conducted every year since 2003. The data have a cross-sectional structure with

independent samples of 6000-7000 individuals in each year. When a household is selected to participate, all household members who are 18 or older are surveyed. The survey has two parts. The household part includes questions on household level variables such as household income and rural versus urban location. The individual part includes questions on happiness, expectations, comparison of current situation to own situation in the past or to the situation of others, and individual level demographic characteristics. The happiness question in the TLSS is the following: "Considering your life as a whole, how happy are you?" The alternative answers are "5: Very happy", "4: Happy", "3: Neutral Neither happy, nor unhappy", "2: Unhappy", "1: Very unhappy". 1 The happiness question in the TLSS is asked at the beginning of the survey, right after some demographic information (such as age, education, marital status and job market status) is collected, lessening concerns about reported happiness being affected by earlier questions in a survey (Kahneman and Krueger, 2006). As shown in Table 1, about 8–9 % is very happy, about 50-54 % is happy, 28 % is neutral in the TLSS data.

Table 1
The descriptive statistics of the variables in the life satisfaction survey (TLSS) (% shares)

Variables		2007	2011
Happiness	Very unhappy	2.2	1.9
	Unhappy	8.8	8.0
	Neutral Neither happy, nor unhappy	28.7	28.0
	Нарру	51.5	53.6
	Very happy	8.7	8.5
Comparison variable	es		'
Economic ladder question (ELQ)	Steps 0–2	14.2	15.6
	Steps 3–4	27.7	32.0
	Step 5	25.9	29.3
	Steps 6–7	23.9	18.2
	Stens 8_10	& 3	<i>1</i> 0

	sichs o-10	د.ن	4.7
Today versus 5 years ago	No opinion	2.2	2.4
	Worse	27.2	23.8
	The same	32.5	33.9
	Better	38.0	39.9
Expectation variables			
Household finances next year	No opinion	11.8	11.2
	Will worsen	11.0	10.1
	Will be the same	53.5	54.4
	Will improve	23.8	24.4
Own working conditions next year	No opinion	11.5	11.7
	Will worsen	8.9	8.0
	Will be the same	52.8	53.7
	Will improve	26.8	26.5
Gender	Female	50.6	51.1
	Male	49.4	48.9
Age group	18–24	16.7	15.4
	25–34	27.1	25.0
	35–44	21.5	21.0
	45–54	16.0	16.8
	55–64	9.9	11.4
	65+	8.9	10.4
Rural/urban Please remove bold	Urban area Please remove bold	63.5 Please remove bold	69.5 Please remove bold
Number of adults in the household	1	7.7	9.7
	2	48.1	47.7

	3	23.5	20.9
	4	13.6	13.7
	5	5.0	
	6	1.6	1.8
	7	0.4	1.1
	8 +	0.1	0.3
Income bracket	Lowest bracket	19.7	13.1
	Second	25.2	24.8
	Third	16.9	24.9
	Fourth	22.6	20.8
	Fifth	12.6	10.9
	Highest bracket	3.1	5.5
Marital status	Never married	18.1	19.0
	Married	74.6	73.1
	Widow/widower	5.5	5.8
	Divorced	1.5	1.7
	Separated	0.3	0.4
Education status	<primary< td=""><td>17.8</td><td>16.6</td></primary<>	17.8	16.6
	Primary school	53.9	53.1
	Secondary school	19.2	18.8
	University	8.5	10.6
	>University	0.6	0.9
Employment status	Employed	39.2	45.7
	Temp. out of work	0.6	0.7
	Unemployed	5.5	4.1
	Housewife	37.0	30.9
	Student	2.9	3.7
	Retired	9.1	8.9

Source Author's calculations. Weights are used. The happiness question is "Considering your life as a whole, how happy are you?". The economic ladder

question is "Think of a well-being ladder on which people living in Turkey stand. Step 0 indicates the lowest level of well-being and step 10 the highest. Where do you see yourself?" The 'today versus 5 years ago' question is "Compared to 5 years ago how do you see your economic situation today?". The expectations questions ask the respondent how she expects the household finances and own working conditions to be next year. The latter question is asked in 2007 only to those who are in the labor market in 2007 and in 2011 only to those who will be in the labor market in 2012

The income brackets in 2007 are: 0–400, 401–600, 601–800, 801–1200, 1201–2500, 2501+ (all in monthly TL). The income brackets in 2011 are: 0–630, 631–990, 991–1650, 1651–2750, 2751–3850, and 3851+ (all in monthly TL)

AQ1

Two comparison variables are used in this study. The first one, ELQ, an external comparison measure, is built on the responses to the 'economic ladder question (ELQ)': "Imagine that people living in Turkey are standing on a well-being ladder where step 0 represents the lowest level and step 10 the highest level of well-being. Where do you see yourself?" About 77–79 % see themselves on steps 3–7, the middle steps (Table 1). Those who see themselves on the top (steps 8–10) are fewer than those who see themselves at the bottom (steps 0–2). The second measure, 'today versus 5 years ago' is an internal comparison measure and it is built on the question "How does your current living standard compare to what you had 5 years ago?" About one-third of the population feels that today is the same as 5 years earlier. A somewhat higher share (38–40 %) feels today is better. Both comparison measures have been used in several studies as indicators of one's self-stated relative position (see, for example, Dumludag 2013 and Senik 2009).

Two expectations variables are used in this study. The first one is based on the question "How do you expect your household income to be next year compared to this year?" The alternative responses are listed as "Will improve", "Will be the same", "Will worsen" and "No opinion". The other question is "How do you expect your own working conditions to be next year?" with the same list of alternative answers. In both questions, a slightly more than half of the population (53–54 %) thought the future will be the same and 11 % had no opinion. About 8–11 % thought things will get worse

and 24–27 % thought things will get better.

The second dataset used in this study, the WVS, aims to investigate sociocultural and political change and to carry out reliable global cross-cultural analyses. A comprehensive and wide-ranging survey of human values, it is conducted by a network of social scientist at leading universities all around world. The survey is made possible by the willingness and capacity of principal investigators in each country to raise local resources and conduct fieldwork using the common questionnaire. Interviews have been carried out with nationally representative samples in more than 80 societies on six continents. Since 1981 a total of six cross-sectional waves of the survey have been carried out, making it possible to carry out analyses of changes over time. The Turkish dataset has about 1300 individuals in 2007 and about 1600 in 2011. More information on technical specifications of the survey can be found at the official website of the survey (http://www.worldvaluessurvey.org/wvs.jsp).²

In the WVS, the happiness variable is based on the answers to the question³: "Taking all things together, would you say you are 4 (very happy), 3 (quite happy), 2 (not very happy), or 1 (not at all happy)". As can be seen in Table 2, 37–38 % of the population is very happy and about 46–49 % is quite happy. These figures are in contrast with the shares in the TLSS according to which less than 10 % is very happy. It appears that the scale of the happiness question matters more than the wording of the categories. Although in both the TLSS and the WVS the highest happiness category is "Very happy", a much smaller share of the respondents pick that category when five categories are available than when four categories are available. We will return to this issue when interpreting the regression estimates.

Table 2

Descriptive statistics of the variables in the world values survey (WVS) (% shares)

	2007	2011
Happiness		
Not at all happy	4.3	3.8
Not very happy	9.4	12.0

c.i roomig		
Quite/rather happy	48.9	46.5
Very happy	37.4	37.7
Life satisfaction		
1: lowest	2.9	2.2
2	1.3	1.7
3	3.6	2.6
4	2.8	3.6
5	5.4	8.2
6	9.9	10.8
7	17.3	19.5
8	21.2	22.7
9	15.1	14.8
10: highest	20.6	14.0
Gender		<u>'</u>
Female	49.8	49.3
Male	50.2	50.7
Age group		
15–24	18.1	13.8
25–34	33.6	27.1
35–44	24.1	25.5
45+	24.3	33.6
Marital status		
Married	66.5	69.2
Div., Sep., Wid. Divorced, Separated, Widowed	3.9	7.8
Single/never married	29.6	23.0
Number of children		
0	35.9	32.0
1	13.6	14.3

3		
2	19.8	23.6
3+	30.8	30.1
Education		
Less than primary	12.1	13.9
Primary school	35.3	30.0
Incomplete secondary	7.2	7.7
Secondary school	26.9	34.1
Incomplete university	7.7	3.2
University or more	10.8	11.1
Type of employment	1	
Self-employed	14.5	9.0
Full time	22.2	25.8
Part time	2.7	5.9
Retired	9.1	12.8
Housewife	34.6	34.6
Student <mark>s</mark>	8.4	6.7
Unemployed	6.1	4.7
Other	2.4	0.5
Social class (subjective)	'	
Upper	2.1	1.7
Upper middle	28.9	22.6
Lower middle	38.0	44.3
Working	25.6	26.4
Lower	5.3	5.0
Income step	1	
1: lowest	22.9	2.2
2	26.6	3.0
3	14.1	8.7

4	14.8	15.9
5	3.2	19.7
6	7.5	16.6
7	1.3	16.9
8	5.8	10.9
9	1.9	5.4
10: highest	1.9	0.6
Health	,	
Poor	8.5	5.7
Fair	23.0	26.2
Good	51.4	48.8
Very good	17.1	19.3
Religion important?	'	
Not at all	2.6	3.0
Not very	6.2	4.0
Rather	16.6	24.7
Very	74.7	68.3
Family important?		
Not at all	0.1	0.1
Not very	0.1	0.1
Rather	2.1	4.3
Very	97.8	95.4
Friends important?	,	·
Not at all	0.8	0.4
Not very	1.9	2.3
Rather	35.5	39.1
Very	61.8	58.2

Leisure important?	-	
Not at all	2.3	1.8
Not very	10.	4 12.2
Rather	40.	1 43.7
Very	47.	3 42.4
Politics important?	<u>'</u>	'
Not at all	32.	1 19.4
Not very	30.	5 32.6
Rather	24.	6 31.7
Very	12.	7 16.3
Work important?	·	
Not at all	4.8	5.8
Not very	8.5	12.1
Rather	30.	6 32.0
Very	56.	2 50.1
Political orientation		
Left	24.	7 19.3
Middle	39.	5 45.7
Right	35.	8 35.0
Trust	·	
Yes	4.8	12.3
No	95.	2 87.7
Proud of nationality	'	,
Not proud	3.6	2.8
Quite proud	15.	1 17.0
Very proud	81.	3 80.2

Source Author's calculations. "Div., Sep., Wid." is "Divorced, Separated, Widowed". The income steps in 2007 are based on absolute monthly incomes 1: 0–500, 2: 500–750, 3: 750–1000, 4: 1000–1250, 5: 1250–1500, 6: 1500–1750, 7: 1750–2000, 8: 2000–2500, 9: 2500–3000, 10: 3000 + all in TI. The income steps

1/30-2000, 0. 2000-2300, 7. 2300-3000, 10. 3000 + , all III 1L. The income steps

in 2011 are self-stated relative positions of individuals on an income ladder. The political orientation variable takes values from 1 (farthest left) to 10 (farthest right). The table shows the shares of those who answer 1–4 (left), 5–7 (middle) and 8–10 (right). The social class variable shows one's subjective view on which social class one belongs to. Trust: Yes (Most people can be trusted), No (Can't be too careful). In 2011, "Proud of nationality" statistics exclude those who are not of Turkish nationality

The life satisfaction variable is based on the answers to the question: "All things considered, how satisfied are you with your life as a whole in these days? Using this card on which 1 means you are "completely dissatisfied" and 10 means you are "completely satisfied" where would you put your satisfaction with your life as a whole?" About 73 % of the population picked 7 or above. The top bracket (10) includes 7–10 times more respondents than the bottom bracket (1). In the WVS data, happiness and life satisfaction are highly correlated; the two variables are not statistically independent.⁴

Other variables in the WVS that are of special interest are political orientation (1: leftmost view, 10: rightmost view) and the importance of several aspects of life such religion, politics, family, friends, work and leisure time. A middle political view (5–7) is the most common view and right view (8–10) is more common than left view (1–4). Almost all respondents (95–98 %) think family is very important. Almost all (97 %) think friends are either rather or very important. Religion is very important to the majority of the Turkish population (68–75 %). Work is very important to about half (50–56 %). In Turkey politics is not deemed to be important in life. About half of the respondents (52–62 %) think politics is either not very or not at all important.

The demographic characteristics that are available in the WVS and TLSS are age and dummy variables for being male, income group, marital status, number of children (only in the WVS), employment status, education status, health (only in the WVS), social class (only in the WVS), rural versus urban location (only in the TLSS). The details of the categorical variables are presented underneath Tables 1 and 2, and the tables that report the regression results.

Since the purpose of this study is to investigate the determinants of happiness and life satisfaction in Turkey in recent years, and since only six waves of the WVS are available, Waves 5 and 6 are selected for the study. These waves were conducted in Turkey in years 2007 and 2011. Consequently, the 2007 and 2011 cross-sections of the TLSS data are used for comparison to the WVS.

As shown in Tables 1 and 2, the gender composition in both survey data is balanced. In both surveys, household income is a categorical variable. In the TLSS, the cutoff points of the income brackets (which can be seen in the notes to Table 1) are determined separately in every survey year. About 85 % of the TLSS sample is in the lowest four income brackets. In the WVS, in 2007 the income question is an absolute one, similar to the income question in the TLSS; but, in 2011 it is a relative one (the self-stated relative position on an imaginary income ladder, similar to the ELQ in the TLSS). When the relative income position is asked, the majority of individuals see themselves somewhere in the middle rather than at the bottom or top.

In both surveys, the majority of the individuals is married, primary school graduate (with only 8 years of completed education), except that in the 2011 WVS the share of secondary school (high school) graduates is higher. Comparing the 2 years, we observe that the 2011 samples in both surveys are somewhat older and better educated than the 2007 samples. Housewives constitute about one-third of the sample. The share of employed individuals is about 40 %.

The WVS has several questions on what is important in life. Almost all Turkish people think family is important in life. About 68–75 % think religion is important. Family and religion are followed by friends, work, leisure and politics. The majority of Turkish people see themselves on the right-wing of the political spectrum.

3.2. Method

In this study, several happiness and life satisfaction equations are estimated where happiness or life satisfaction of an individual i is expressed as a function of the demographic and economic characteristics of the individual.

The two equations can be written as follows:

$$Happiness_i = \alpha + \beta D_i + \varepsilon_i.$$

$$Life\ Satisfaction_i = \gamma + \theta D_i + u_i.$$

In the above, *Happiness* takes values between 1 and 5 in the TLSS and between 1 and 4 in the WVS. *Life Satisfaction* takes values between 1 and 10 (asked only in the WVS), α and γ are the constants, β and θ are the coefficients of the control variables, the D_i matrix includes the control variables in the regressions, ε_i and u_i are the random error terms.

In this study the ordered logit model is used to estimate the happiness and life satisfaction equations, since it is a model that takes into account that the dependent variables are discrete and ordered variables. To show the effect of a change in the value of the control variables on the probability of being very happy or having the highest level of satisfaction (in the WVS), the marginal effects are estimated for the highest level of the outcome variable using the coefficient estimates of the ordered logit model.

The specifications of the regression equations and the definitions of the control variables are based on the practice in the literature and on the survey questions. For example, to see whether the U-shaped relationship between age and happiness is valid, "age" and "age squared" are used in one set of regressions. To compare the findings to the results of the earlier studies that use the WVS, the age group definition in those studies is adopted. Another example is that the education variables in the TLSS and WVS regressions are slightly different, because the categories of education are defined differently in the two surveys (see Tables 1 and 2).

4. Results

In this section, first, the estimates obtained from the two surveys are described in detail. It is seen that many of the control variables are statistically significantly correlated with happiness; however, the level of statistical significance is usually lower for the estimates obtained using the WVS data because of the smaller sample. Many results (both signs and

magnitudes of the estimates) are consistent across the two surveys, but some are not. One reason for inconsistent findings may be that the set of control variables is different across surveys. Another reason may be that the variables are defined differently. Therefore, comparison across surveys must be done properly. The second task done in this section is to re-estimate regressions by restricting attention to the set of control variables that are common to both surveys and by defining the control variables in the same manner.

The tables in this section present the marginal effects of the ordered logit model (computed at the sample average, for being at the highest category of happiness or life satisfaction) and their standard errors. Tables 3 and 4 show the estimates based on the TLSS data in years 2007 and 2011. In columns (1) and (2) the effects of the two comparison variables are shown; in columns (3) and (4) the effects of the two expectations variables are shown. In column (5), the regression includes a comparison and an expectation variable to show the relative effects of the two. Tables 5 and 6 show the estimates based on using the WVS data in years 2007 and 2011. In the WVS both happiness and life satisfaction are asked. In columns (1) and (2) the dependent variable is happiness; in columns (3) and (4) the dependent variable is life satisfaction. Results on life satisfaction are reported for comparison with the literature. They are not meant to be related to the results on happiness. Since the two concepts are different, regression estimates are not directly comparable.

Table 3Estimates of the happiness equation (TLSS 2007 data): Marginal effects for the probabi and statistical significance

	Comparisons				Expectations		
(1)		(2)		(3)			
ME	SE	ME	SE	ME	SE		
-0.007	0.001***	-0.005	0.001***	-0.006	0.001***		
0.000	0.000***	0.000	0.000***	0.000	0.000***		
-0.009	0.006***	-0.011	0.006*	-0.011	0.006*		
'							
	ME -0.007 0.000	ME SE -0.007 0.001*** 0.000 0.000***	ME SE ME -0.007 0.001*** -0.005 0.000 0.000*** 0.000	ME SE ME SE -0.007 0.001*** -0.005 0.001*** 0.000 0.000*** 0.000 0.000***	ME SE ME SE ME -0.007 0.001*** -0.005 0.001*** -0.006 0.000 0.000*** 0.000 0.000*** 0.000		

			e.Frooming			
2nd Income bracket	0.019	0.004***	0.024	0.004***	0.022	0.004***
3rd Income bracket	0.029	0.006***	0.043	0.006***	0.041	0.006***
4th Income bracket	0.032	0.005***	0.043	0.005***	0.042	0.006***
5th Income bracket	0.074	0.009***	0.091	0.009***	0.083	0.009***
6th Income bracket	0.092	0.019***	0.124	0.021***	0.123	0.022***
Marital status						
Married	0.041	0.006***	0.042	0.006***	0.039	0.006***
Widow/widower	-0.007	0.007***	-0.005	0.008	-0.011	0.008
Divorced	-0.024	0.008***	-0.027	0.007***	-0.027	0.008***
Separated	-0.040	0.008***	-0.040	0.008***	-0.049	0.007***
Education						
Primary school	-0.002	0.006	0.003	0.006	0.006	0.006
High school	-0.007	0.008	0.005	0.008	0.014	0.008*
University	-0.018	0.009***	-0.003	0.009	0.002	0.009
> University	-0.049	0.014***	-0.040	0.015***	-0.029	0.018*
Unemployed	-0.041	0.009***	-0.042	0.009***	-0.052	0.009***
Temp.out of work	-0.038	0.023***	-0.027	0.024	-0.032	0.026
Housewife	0.001	0.006	0.004	0.006	0.001	0.007
Student	-0.006	0.012	0.010	0.012	0.007	0.013
Retired	-0.012	0.007***	-0.008	0.007	-0.008	0.008
Urban	-0.008	0.004***	-0.012	0.004***	-0.013	0.004***
Number of adults		-		-		
2	0.005	0.007	0.003	0.008	0.012	0.008
3	0.002	0.008	0.003	0.008	0.010	0.008
4	0.009	0.009	0.007	0.009	0.012	0.009

			e.Proofing			
5	-0.009	0.010	-0.012	0.010	0.000	0.011
6	0.027	0.019***	0.025	0.019	0.037	0.020*
7	0.149	0.064***	0.160	0.068**	0.150	0.068**
8+	-0.045	0.032***	-0.060	0.022***	-0.050	0.024**
Comparison varia	bles		'		'	
ELQ steps						
0–2	-0.083	0.007***				
3–4	-0.032	0.005***				
6–7	0.019	0.005***				
8–10	0.071	0.008***				
Today versus 5 ye	ars ago					
Worse			-0.031	0.004***		
Better			0.039	0.005***		
Expectation variate	bles					
Household income	2					
Lower					-0.042	0.003***
Higher					0.056	0.006***
Own working con-	ditions					
Worse						
Better						
N	6442		6376		5787	
Log likelihood	-7328.8		-7319.7		-6602.9	
Pseudo R2	0.999		0.999		0.999	

Source Author's calculations. "ME" and "SE" denote marginal effects and standard e stars indicate whether the marginal effects are statistically significantly different fron p < 0.01. The base categories are female, the lowest income bracket, never married, i one-adult household, the middle step of the economic ladder, and no opinion (today v household income and own working conditions). Primary school represents 8 years o are: 0-400, 401-600, 601-800, 801-1200, 1201-2500, 2501+ (all in monthly TL). The 631-990, 991-1650, 1651-2750, 2751-3850, and 3851+ (all in monthly TL). In the labor market related variables cannot all be estimated because the question on expect asked only to those who are employed or temporarily out of work

Table 4Estimates of the happiness equation (TLSS 2011 data): Marginal effects for the probastatistical significance

	Comparisons				Expectations	
	(1)	1) (2)			(3)	
	ME	SE	ME	SE	ME	SE
Age	-0.007	0.001***	-0.007	0.001***	-0.009	0.001***
Age squared	0.000	0.000***	0.000	0.000***	0.000	0.000***
Male	-0.005	0.005***	-0.007	0.005	-0.007	0.005
Income						
2nd Income bracket	0.009	0.004***	0.013	0.004***	0.015	0.005***
3rd Income bracket	0.027	0.005***	0.031	0.005***	0.035	0.005***
4th Income bracket	0.041	0.006***	0.048	0.006***	0.057	0.007***
5th Income bracket	0.059	0.011***	0.079	0.012***	0.088	0.012***
6th Income bracket	0.073	0.014***	0.098	0.015***	0.113	0.016***
Marital status						
Married	0.051	0.005***	0.054	0.005***	0.060	0.005***
Widow/widower	0.011	0.007***	0.009	0.007	0.007	0.008
Divorced	-0.025	0.005***	-0.021	0.006***	-0.022	0.006***
Separated	-0.029	0.008***	-0.027	0.009***	-0.030	0.008***

Education						
Primary school	0.008	0.005***	0.017	0.005***	0.023	0.005***
High school	0.005	0.007	0.020	0.007***	0.021	0.007***
University	0.007	0.008	0.027	0.009***	0.026	0.009***
> University	-0.019	0.016***	0.006	0.019	0.004	0.018
Unemployed	-0.042	0.009***	-0.037	0.009***	-0.049	0.010***
Temp.out of work	-0.007	0.020	-0.012	0.020	-0.009	0.021
Housewife	0.024	0.005***	0.024	0.006***	0.022	0.006***
Student	0.002	0.010	0.004	0.011	0.004	0.011
Retired	0.024	0.007***	0.022	0.007***	0.023	0.008***
Urban	-0.025	0.004***	-0.027	0.004***	-0.029	0.005***
Number of adults						
2	0.012	0.006***	0.008	0.006	0.007	0.007
3	0.014	0.007***	0.014	0.007**	0.011	0.008
4	0.005	0.007	0.003	0.008	-0.005	0.008
5	-0.003	0.009	0.003	0.010	-0.001	0.010
6	-0.007	0.013	0.006	0.015	-0.012	0.014
7	-0.016	0.013***	-0.028	0.012**	-0.029	0.015*
8+	-0.051	0.012***	-0.046	0.015***	-0.051	0.018***
Comparison varia	ibles					
ELQ steps						
0–2	-0.063	0.006***				
3–4	-0.031	0.005***				
6–7	0.020	0.005***				
8–10	0.070	0.009***				
Today versus 5 ye	ears ago					
Worse			-0.031	0.003***		

		e.Proofing			
Better		0.045	0.005***		
Expectation vario	ables				
Household incom	ne				
Lower				-0.041	0.003
Higher				0.058	0.006
Own working con	nditions				
Worse					
Better					
N	7361	7190		6530	
Log likelihood	-8184.5	-7969.8		-7171.8	
Pseudo R2	0.900	0.900		0.900	

Table 5

Estimates of the happiness and life satisfaction equations (WVS 2007 data): Margir probability of being very happy or having the highest level of life satisfaction, their statistical significance

	Happin	Happiness				Life satisfaction			
	(1)	(1)		(2)		(3)			
	ME	SE	ME	SE	ME	SE	MF		
Male	-0.084	0.040**	-0.066	0.040*	0.010	0.026	0.0		
Age group	'				'				
25–34	-0.043	0.045	-0.027	0.045	-0.003	0.032	0.0		
35–44	-0.038	0.055	-0.015	0.055	-0.050	0.037*	-0.		
45+	-0.061	0.060	-0.021	0.062	-0.065	0.040*	-0.		
Marital statu	S				'				
Divorced, Separated, Widowed	-0.091	0.066*	-0.101	0.065*	-0.070	0.039**	-0.		

-0.0
<u> </u>
-0.0
-0.0
0.0
0.00
-0.0
-0.0
-0.0
0.03
0.00
-0.0
0.0
¢ 0.0′
0.03
** -0.0
-0.0
<u> </u>
*** 0.00
*** 0.12
*** 0.38
0.00
*

-0.106	0.077*	0.111	00-61	0.040		
	0.077	-0.111	0.076*	-0.040	0.055	-0.
0.003	0.023	-0.019	0.023	0.016	0.015	0.0
0.035	0.018**	0.044	0.019***	0.028	0.012**	0.02
-0.009	0.013	-0.005	0.013	-0.003	0.009	-0.0
0.028	0.017*	0.033	0.018**	0.011	0.011	0.0
0.012	0.005**	0.011	0.006**	0.004	0.004	0.00
0.110	0.057**	0.138	0.057***	0.061	0.037**	0.00
0.057	0.025**	0.060	0.025***	0.055	0.017***	0.03
			,		'	
0.013	0.039			0.033	0.026*	
0.001	0.048			0.052	0.033*	
0.025	0.046			0.031	0.030	
-0.053	0.076			-0.001	0.050	
-0.120	0.052**			-0.018	0.033	
-0.204	0.083***			-0.093	0.048**	
-0.081	0.058*			-0.006	0.037	
-0.120	0.082*			0.023	0.062	
0.116	0.107			0.040	0.066	
		0.124	0.114			0.04
		0.087	0.061*			0.08
	0.035 -0.009 0.028 0.012 0.110 0.057 0.013 0.001 0.025 -0.053 -0.120 -0.204 -0.081 -0.120	0.035 0.018** -0.009 0.013 0.028 0.017* 0.012 0.005** 0.057 0.025** 0.013 0.039 0.001 0.048 0.025 0.046 -0.053 0.076 -0.120 0.052** -0.081 0.058* -0.120 0.082*	0.035 0.018** 0.044 -0.009 0.013 -0.005 0.028 0.017* 0.033 0.012 0.005** 0.011 0.110 0.057** 0.138 0.057 0.025** 0.060 0.013 0.039 0.048 0.025 0.046 0.048 -0.053 0.076 0.052** -0.120 0.052** 0.083*** -0.081 0.058* 0.116 0.116 0.107 0.124	0.035 0.018** 0.044 0.019*** -0.009 0.013 -0.005 0.013 0.028 0.017* 0.033 0.018** 0.012 0.005** 0.011 0.006** 0.110 0.057** 0.138 0.057*** 0.057 0.025** 0.060 0.025*** 0.013 0.039 0.001 0.048 0.025 0.046 0.052** -0.120 0.052** 0.052** -0.204 0.083*** 0.058* -0.120 0.082* 0.116 0.116 0.107	0.035 0.018** 0.044 0.019*** 0.028 -0.009 0.013 -0.005 0.013 -0.003 0.028 0.017* 0.033 0.018** 0.011 0.012 0.005** 0.011 0.006** 0.004 0.110 0.057** 0.138 0.057*** 0.061 0.057 0.025** 0.060 0.025*** 0.055 0.013 0.039 0.033 0.001 0.048 0.052 0.025 0.046 0.031 -0.053 0.076 -0.001 -0.120 0.052** -0.018 -0.204 0.083*** -0.093 -0.081 0.058* -0.006 -0.120 0.082* 0.023 0.116 0.107 0.040	0.035 0.018** 0.044 0.019*** 0.028 0.012** -0.009 0.013 -0.005 0.013 -0.003 0.009 0.028 0.017* 0.033 0.018** 0.011 0.011 0.012 0.005** 0.011 0.006** 0.004 0.004 0.110 0.057** 0.138 0.057*** 0.061 0.037** 0.057 0.025** 0.060 0.025*** 0.055 0.017*** 0.013 0.039 0.033 0.026* 0.001 0.048 0.031 0.030 0.025 0.046 0.031 0.030 -0.053 0.076 -0.001 0.050 -0.120 0.052** -0.018 0.033 -0.204 0.083*** -0.093 0.048** -0.010 0.082* 0.023 0.062 0.116 0.107 0.040 0.066

class		0.039	0.058		0.03
Upper class		0.091	0.060*		0.0
N	1019	983		1019	983
Log likelihood	-950.9	-919.1		-1919.2	-18
Pseudo R2	0.11	0.107		0.056	0.0:

Source Author's calculations. ME" and "SE" denote marginal effects and standard er regressions are weighted. The stars indicate whether the marginal effects are statistical significantly different from zero. * p < 0.10, ** p < 0.05, *** p < 0.01. The categorie educational attainment are primary school graduate (8 years of education), incomplet school (high school), high school graduate, incomplete university and university grad groups are described in the notes to Table 2. The base categories in the regressions a group, married, no children, less than primary school graduate, self-employed, poor by think that religion/family/friends/leisure time/politics/work is important, the farthest view, does not think that most people can be trusted, not proud of nationality, lowest (1 out of 10) and lower social class

Table 6

Estimates of the happiness and life satisfaction equations (WVS 2011 data): Mar probability of being very happy or having the highest level of life satisfaction, their statistical significance

	Happine	ess			Life satisfaction		
	(1)		(2)		(3)		
	ME	SE	ME	SE	ME	SE	
Male	0.018	0.034	0.017	0.034	-0.011	0.018	
Age group					<u>'</u>		
25–34	-0.003	0.049	0.004	0.050	-0.024	0.028	
35–44	-0.042	0.054	-0.042	0.054	-0.040	0.030*	
45+	0.060	0.059	0.067	0.059	-0.010	0.033	
Marital status			'			'	
Divorced, separated, widowed	-0.069	0.041**	-0.070	0.041**	-0.003	0.023	

Single/never married	-0.056	0.047	-0.061	0.047*	-0.032	0.022*
Number of chi	ldren					
1	0.012	0.046	0.004	0.046	0.004	0.024
2	0.015	0.044	0.017	0.045	0.006	0.022
3+	0.059	0.046	0.056	0.047	0.010	0.024
Education						
Primary school	0.024	0.039	0.006	0.039	-0.046	0.024**
Incomplete secondary	0.029	0.055	0.001	0.056	-0.063	0.030**
Secondary school	0.034	0.043	0.011	0.045	-0.047	0.027**
Incomplete university	0.100	0.086	0.044	0.086	0.007	0.053
University or more	0.081	0.055*	0.063	0.057	-0.061	0.030**
Type of emplo	yment					
Full time	0.000	0.045	-0.015	0.046	0.024	0.023
Part time	0.016	0.061	0.011	0.063	0.041	0.034
Retired	-0.055	0.052	-0.077	0.053*	-0.022	0.025
Housewife	0.029	0.056	0.020	0.057	0.015	0.028
Students	-0.021	0.076	-0.008	0.079	-0.008	0.037
Unemployed	-0.067	0.067	-0.075	0.070	-0.053	0.029**
Other	-0.227	0.096***	-0.228	0.104**	-0.100	0.031***
Health						
Fair	0.193	0.023***	0.194	0.023***	0.046	0.010***
Good	0.315	0.021***	0.322	0.021***	0.099	0.012***
Very good	0.617	0.033***	0.624	0.033***	0.264	0.026***
Importance of religion	0.050	0.018***	0.047	0.018***	0.019	0.009**

Importance of family	0.090	0.049**	0.092	0.051**	0.024	0.025	(
Importance of friends	-0.024	0.022	-0.019	0.022	-0.004	0.011	(
Importance of leisure time	0.029	0.017**	0.025	0.017*	0.014	0.009**	(
Importance of politics	-0.011	0.012	-0.005	0.012	0.000	0.006	(
Importance of work	-0.004	0.015	-0.003	0.015	0.006	0.007	(
Political orientation (left–right)	0.009	0.005**	0.008	0.005**	0.007	0.003***	(
Trust to others	0.077	0.034**	0.077	0.034**	0.031	0.017**	(
Proud of nationality	0.073	0.024***	0.069	0.025***	0.022	0.012**	,
Income scale				'		'	
2	0.260	0.090***		0.147	0.043***		
3	0.229	0.068***		0.099	0.024***		
4	0.143	0.061***		0.070	0.017***		
5	0.158	0.060***		0.080	0.017***		Γ
6	0.158	0.061***		0.108	0.019***		
7	0.201	0.062***		0.136	0.021***		Γ
8	0.243	0.065***		0.191	0.027***		Γ
9	0.253	0.078***		0.217	0.040***		Γ
10	0.475	0.163***		0.273	0.112***		Г
Social class							
Working class			0.193	0.109**			(
Lower middle class			0.165	0.059***			(
Upper			0.097	0.055**			(

middle class					
Upper class		0.113	0.056**		1
N	1202	1205		1202	
Log likelihood	-1163.7	-1170.9		-2274.9	
Pseudo R2	0.111	0.107		0.053	-(

Source Author's calculations

See notes to Table 5

The U-shaped relationship between age and happiness is well documented in the literature. Usually, the age of lowest happiness is found to be between 45 and 50 (Graham 2010). Calculations based on the estimates of marginal effects of age and age squared variables indicate that, in Turkey, the age at which the lowest level of happiness is experienced is in the lower 50s. The two earlier studies on Turkey introduce age as a categorical variable; therefore, estimating the age of lowest happiness is not possible based on their results (Selim 2008; Dumludag 2013).

Males are less happy than females in both years, according to the TLSS results. The WVS data yield the same finding only in year 2007. In 2011, no gender difference in happiness is detected. No gender difference exists in life satisfaction. Selim (2008) finds that women are happier, but Dumludag (2013) finds no difference in happiness between the two sexes. In other countries, results vary. Women have been found to be happier in Western Europe and the United States, whereas in Central and Eastern Europe and in Latin America men are happier (Caporale et al. 2009; Graham and Pettinato 2002; Guriev and Zhuravskaya 2009; Graham 2010).

In the literature, income is found to be an important determinant of happiness (Easterlin 1974; Graham 2010). Studies on Turkey find positive and statistically significant effect of income on happiness and life satisfaction (Selim 2008; Dumludag 2013). My findings on the effect of income vary depending on the definition of income (absolute or relative) and on the survey

data used. In the TLSS sample, those in the higher income brackets (instead of being in the lowest bracket) are usually more likely to be "very happy".⁵ The estimates indicate that those who are on the highest income bracket are happier by about 0.07–0.13 points. On a 1–5 scale, the difference is nonnegligible, thus economically significant (Tables 3, 4). In the 2007 WVS data (Table 5), compared to being in the lowest income group, being in the upper middle income groups (6, 7, 8, and 9) is correlated with less happiness (for which the marginal effects are statistically significant) and less life satisfaction (for which the degree of significance is smaller). Middle income groups are less happy than the poorest. Such a finding is surprising indeed. In the 2011 WVS data, in which the income groups are relative and not absolute, seeing self on a higher income scale relative to others is correlated with greater happiness and life satisfaction (for which both marginal effects are statistically significant) (Table 6). Thus, the estimates support the wellknown finding in the literature that relative income is an important determinant of well-being (and sometimes an even more important determinant than absolute income) (see for example, Stark and Taylor 1991; Clark and Oswald 1998; Frey and Stutzer 2002; Senik 2004; Ferrer-i-Carbonell 2005; Layard 2005; Senik 2009; Layard et al. 2010).

The marital status of an individual is strongly correlated with happiness. In the TLSS, I find that compared to those who are never married, married individuals are happier and divorced or separated individuals are less happy. In the WVS, the overall finding is that, compared to those who are married, never married individuals and divorced, separated or widowed individuals are less happy and have lower life satisfaction. In the literature, similar findings are reported (Graham 2010; Selim 2008); however, Dumludag (2013) found no significant effect for Turkey. Therefore the findings are not robust across different datasets and times.

The results about the effect of education on happiness depend on the dataset used and on the time analyzed. In the TLSS 2007 data, more education is not associated with greater happiness; those with the highest level of education (more than a university degree) are indeed less likely to be very happy compared to those with no educational degree. In the TLSS 2011 data more education is good for happiness, but not for those with the highest level of education (again compared to those with no degree). The WVS data yield

conflicting results, too. In 2007, those with more education are found to be less happy; whereas in 2011 no significant effect is detected. When the explained variable is life satisfaction, the WVS results are consistent across time: in both survey years, more education is associated with a lower probability of being very happy.

Such findings on the effects of education show us that it is ambiguous whether educational attainment makes us any happier and that the question needs to be studied further. In fact, conflicting results are not uncommon in the literature. One factor that generates different results may be the lack of a consistent definition of educational categories. For example, Selim (2008) somewhat arbitrarily categorizes the education variable in the WVS data into lower, middle and upper levels and finds that having a middle level of education reduces happiness. Another factor may be that education has both negative and positive effects on happiness and that the net effect depends on the exact set of economic circumstances. For example, some studies report that any increase in the level of educational attainment translates into higher happiness [in England and in the US (Blanchflower and Oswald 2004)]. Others find that the highest happiness is achieved at the middle level of education [in Switzerland (Stutzer 2004)]. Some find that the positive effect of education on happiness is stronger in low-income countries than in highincome countries (Fahey and Smyth 2004; Ferrer-i-Carbonell 2005). Another One study detects no significant relationship between the two variables (Flouri 2004). Another factor that complicates the link between education and happiness/life satisfaction is the direction of causality: It may be the case that more education is achieved by those who feel happier. It may also be the case that an omitted variable, such as personality, influence both the educational attainment and happiness/life satisfaction. Such concerns can be addressed by the help of an exogenous shock to one variable to see the effect on the other variable. An unanticipated education reform is one example.

One's labor market status is strongly correlated with both happiness and life satisfaction. The omitted categories in the two datasets are different: "employed" in the TLSS and "self-employed" in the WVS. The difference is tolerated for the sake of replicating the earlier studies (for example, Graham 2010; Selim 2008). I find that being unemployed is associated with lower

happiness and lower life satisfaction compared to being employed (in the TLSS) or being self-employed (in the WVS), with higher degrees of statistical significance found for the estimates in the TLSS. Earlier studies on Turkey have found similar results. Dumludag (2013) reported that those who worked for income in the past 12 months are happier on average than those who did not work. Selim (2008) found in most regressions that the unemployed individuals are less happy than the self-employed. In the 2011 TLSS and 2007 WVS, housewives are found to be happier than the employed individuals, with no significant effect in the other samples.

People living in rural areas (i.e. areas with a population of 20,000 or less) are happier on average than those living in urban areas (TLSS regressions). The estimates yield no clear information on the relationship between happiness and household size, except that those who live in households with 8 or more adults are less happy in both years of the TLSS. No clear evidence emerges either on the effects of number of children (Tables 5, 6, WVS data).

Health status has a strong correlation with well-being. The WVS asks the respondents to evaluate their own health (poor, fair, good, very good). The estimates in Tables 5 and 6 show that better health is always associated with both higher happiness and higher life satisfaction; however, the finding does not indicate causality. As in the case of educational attainment, causality may run backwards or an omitted third variable may determine both of the variables.

In the TLSS, we have information on comparisons (how one compares his own income to a benchmark: others' income or to his own income in the past) and expectations (how one expects his living conditions, mainly income, and working conditions to be in the next year, the same as the current year, worse or better). On the effects of comparison and expectation variables, the main findings are that favorable comparisons and expectations are associated with higher happiness and unfavorable ones are associated with lower happiness (Tables 3, 4). Consistent with Dumludag (2013) and Senik (2009) who use Life in Transition Survey, and Caner (2014) who uses the TLSS, ranking high on the societal well-being ladder has a positive impact on individual life satisfaction and vice versa. The negative effect on happiness of seeing self on the lowest steps of the societal well-being ladder is so large that it is similar

to falling down from the fifth income bracket to the lowest one. The aforementioned studies suggest that comparisons are asymmetric, but the estimates obtained here yield no support for asymmetry. (A hypothesis test of beta[ELQ 0-2] + beta[ELQ 8-10] = 0 has a p value of 0.27. A similar hypothesis test for 'today versus 5 years ago' has a p value of 0.25.)

An expectation of a higher income next year makes one happier, relative to someone who expects a lower income or the same income as the current year (Tables 3, 4). In fact, the direction of the association is theoretically ambiguous. An expectation of higher income in the future may reflect an aspiration for a better life or a forecast of higher income, or both, thus making current income seem worse. For a given income level, higher income aspirations (Stutzer 2004) or higher expected future income (Van Praag and Ferrer-i Carbonell 2008) reduce current utility. On the other hand, if expectations reflect a forecast of higher income, expected future income and current happiness are positively correlated. Here, income expectations are seen as a consumption good (Frijters et al. 2012), unlike the utility function in traditional economic theory with only current consumption as an argument.

The effect on happiness of expecting higher income is similar to the effect of being in the 4th or 5th income bracket instead of the lowest bracket; in other words it is quite large (Tables 3, 4). The second expectation variable, expecting better working conditions, influences happiness in the same direction and at about the same magnitude. There are a few studies in the literature that have attempted to quantify the effect of expected future income on happiness. Knight et al. (2009) find that a higher expected rise in income is associated with higher subjective well-being among China's rural-urban migrants, and that the estimated effect of expectations is well in excess of the possible effect of higher income. Gao and Smyth (2011) find that happiness generated by an expectation of a big increase in income is equivalent to a 380 % rise in average monthly income. Another study on China found that in the urban, migrant, and rural samples an income increase of 85 %, 420 % and an astonishing 1400 %, respectively, would be equivalent to a change in expectations from neutral to significant improvement (Frijters et al. 2012). Using the TLSS 2010–2011 data, Caner (2014) estimated that expecting higher income generates an increase in happiness that would be generated by a roughly 320 % increase in income. Apparently, expecting a better future has

a sizable positive effect on today's happiness, but the estimated magnitude of the effect is smaller in Turkey than it is in China.

To gauge the relative effects of comparison and expectation variables, a regression which includes both a comparison variable (ELQ) and an expectation variable (expectations about household income in the following year) is estimated. The estimates can be seen in column (5) in Tables 3 and 4. Evidently, the ELQ variable has a stronger effect on happiness than the expectation variable. We can quite confidently say that being on steps 0-2 has a stronger negative effect on happiness than expecting income to be lower. For example, in 2007, the hypothesis of beta[ELQ 0-2] = beta[Expect lower income] is rejected with a Chi square test statistic of 5.73 and a p value of 0.017.

In the WVS, we have information on the respondents' views about what are important in life for them. Although most results vary according to survey year, one consistent finding is that those who value leisure time more are on average happier and they have higher life satisfaction. Maybe they simply know how to enjoy life. An alternative explanation is that causality runs from happiness to giving priority to leisure time in life. Those who think that religion is important in life are happier than others, but the finding is only valid for 2011. No consistent findings are found for how other aspects of life, such as friends, family, politics, or work are related to happiness and life satisfaction. The results are not consistent with the results of Selim (2008) who uses the same survey data in an earlier year and reports that those who think religion, family, and friends are important are happier, whereas those who think politics and work are important are less happy.

The estimates show that the respondents with a rightward political orientation are happier than those with a leftward orientation; furthermore, they have higher life satisfaction in 2011, exactly the same result as found in 1990, 1996 and 2001 waves of the WVS (Selim 2008). It may, of course, be that those with a rightward political orientation have benefitted more from the economic and political environment that exists in Turkey since 2002.

Other variables that might be associated with one's happiness are indicators of how well the person fits in the society that he lives in. Those who trust

others are happier and have higher life satisfaction than those who do not trust others. So are those who are proud of their nationality. Another indicator of one's social position is the social class he thinks he belongs to. Relative to those who think they belong to the lower class, those who are in the working class, lower middle class, upper middle class, or the upper class have higher happiness and life satisfaction, but only in the 2011 sample of the WVS (Tables 5, 6).

4.1. Comparison Across Datasets, Using the Same Set of Control Variables

If the set of control variables were constructed in the same manner and the variables were similarly defined in the WVS and the TLSS, would the two datasets generate similar descriptions of the characteristics of happy people? There are, of course, variables that are omitted from the regressions. Some are individual fixed effects (unobserved heterogeneity), others are omitted time-varying variables. In the absence of these, the equations are not fully specified. Regardless of the omitted variables, with such large samples, —one would expect the bias potentially generated by these omitted variables to be similar across the two surveys given the large sample size, thereby generating similar estimates for the coefficients.

The estimations are redone with the same set of control variables that are similarly defined. Tables 7 and 8 show us that matching the set of control variables and the variable definitions are not enough to obtain estimates that are similar in direction and magnitude. Although there are similarities, the WVS and the TLSS samples yield many different results. The details are explained below.

Table 7 Tables 7 and 8 are missing. Only half of the results are shown. Please see the attached file for the full table.

Comparison of the estimates across the two datasets (WVS and TLSS, 2007): Marginal effects for the probability of being very happy, their standard errors and statistical significance

	ME	SE	Robust SE
World values survey (WVS)			
Age	-0.021	0.006***	0.006***

0-		e.Proofing	
Age squared	0.000	0.000***	0.000***
Male	-0.005	0.038	0.041
Income group		<u>'</u>	
2	0.034	0.036	0.038
3	0.049	0.044	0.047
4	0.027	0.043	0.043
5	-0.043	0.042	0.043
6	-0.036	0.069	0.065
Marital status	<u>'</u>	<u>'</u>	
Married	0.113	0.036***	0.037***
Widow/widower	-0.062	0.073	0.078
Divorced	-0.082	0.080	0.075
Separated	-0.155	0.185	0.248
Education	'	'	'
Primary school	0.061	0.041	0.045
High school	0.032	0.048	0.051
University	0.079	0.062	0.065
>University	0.432	0.220**	0.210**
Unemployed	-0.129	0.054***	0.057**
Housewife	0.017	0.045	0.049
Student	0.007	0.063	0.055
Retired	-0.020	0.052	0.056
N	1318		1318
Log likelihood	-1393.1		-1394.8
Pseudo R2	0.019		0.018
Turkish life satisfaction	survey (TLSS)		
Age	-0.013	0.001***	0.002***

Age squared	0.000	0.000***	0.000***
Male	-0.024	0.011**	0.013*
Income group			
2	0.085	0.014***	0.016***
3	0.134	0.014***	0.017***
4	0.135	0.014***	0.017***
5	0.187	0.013***	0.016***
6	0.197	0.013***	0.015***
Marital status			
Married	0.094	0.014***	0.017***
Widow/widower	-0.040	0.027*	0.030
Divorced	-0.121	0.037***	0.039***
Separated	-0.254	0.056***	0.047***
Education			
Primary school	0.006	0.011	0.013
High school	0.009	0.014	0.017
University	-0.006	0.018	0.022
>University	-0.094	0.061*	0.077
Unemployed	-0.091	0.016***	0.024***
Housewife	0.005	0.012	0.014
Student	0.017	0.023	0.034
Retired	-0.017	0.014	0.015
N	6442		6442
Log likelihood	-7546.9		-7546.9
Pseudo R2	0.900		0.900

Source Author's calculations. "ME" and "SE" denote marginal effects and standard errors. The ten brackets of monthly household income in the WVS are 0–500, 500–750, 750–1000, 1000–1250, 1250–1500, 1500–1750, 1750–2000, 2000–2500, 2500–3000, 3000+. The six brackets of monthly household income in the TLSS are 0–400, 401–600, 601–800, 801–1200, 1201–2500, 2501+. Some income brackets in the WVS are merged to match the TLSS brackets as closely as

possible. The education categories in the WVS are redefined (incomplete degrees

are merged with the lower degree) to match those in the TLSS. The base categories are female, the lowest income bracket, never married, illiterate or no degree, and employed. Primary school represents 8 years of education. * p < 0.10, ** p < 0.05, *** p < 0.01

Table 8 Tables 7 and 8 are missing. Only half of the results are shown. Please see the attached file for the full table.

Comparison of the estimates across the two datasets (WVS and TLSS, 2011): Marginal effects for the probability of being very happy, their standard errors and statistical significance

	ME	SE	Robust SE
World values survey (W	VVS)		
Age	-0.006	0.005	0.006
Age squared	0.000	0.000	0.000
Male	0.013	0.034	0.035
ncome group	'	'	<u>'</u>
2	0.225	0.083***	0.114**
3	0.148	0.059***	0.078*
4	0.149	0.055***	0.074**
5	0.204	0.055***	0.074***
6	0.209	0.056***	0.075***
7	0.220	0.056***	0.074***
8	0.266	0.061***	0.080***
9	0.322	0.071***	0.083***
10	0.517	0.149***	0.142***
Marital status			
Married	0.093	0.034***	0.034***
Widow/widower	0.015	0.064	0.069
Divorced	-0.029	0.066	0.074
Senarated	-0.275	0 036***	0 038***

Separateu	0.413	e.Proofing	0.030
Education		·	
Primary school	0.037	0.037	0.044
High school	0.013	0.040	0.050
University	0.063	0.052	0.058
Unemployed	-0.136	0.054***	0.057**
Housewife	0.003	0.041	0.043
Student	-0.014	0.058	0.057
Retired	-0.075	0.043**	0.048
N	1565		1565
Log likelihood	-1690.2		-1690.7
Pseudo R2	0.021		0.021
Turkish life satisfaction	survey (TLSS)	<u>'</u>	'
Age	-0.012	0.001***	0.001***
Age squared	0.000	0.000***	0.000***
Male	-0.014	0.008*	0.009
Economic ladder questi	on (ELQ)	'	
1	0.043	0.035	0.048
2	0.129	0.031***	0.043***
3	0.135	0.030***	0.041***
4	0.202	0.029***	0.040***
5	0.243	0.028***	0.040***
6	0.272	0.029***	0.040***
7	0.283	0.029***	0.040***
8	0.298	0.028***	0.040***
9	0.299	0.028***	0.040***
		0.033***	0.048***

Married	0.116	0.013***	0.016***
Widow/widower	0.010	0.024	0.026
Divorced	-0.153	0.033***	0.037***
Separated	-0.191	0.062***	0.063***
Education			
Primary school	0.024	0.011***	0.013*
High school	0.029	0.014***	0.016*
University	0.041	0.014***	0.017**
Unemployed	-0.093	0.016***	0.020***
Housewife	0.034	0.009***	0.010***
Student	-0.008	0.019	0.024
Retired	0.042	0.013***	0.013***
N	7368		7368
Log likelihood	-8222.4		-8223.1
Pseudo R2	0.900		0.900

Source Author's calculations

The income brackets in the WVS show self-stated relative income; therefore, the ELQ variable is used in the TLSS. The education categories in both datasets are redefined

The similarities in the two sets of estimates are that the unemployed are less happy than others; married individuals are happier than singles; divorced, separated or widowed individuals are less happy than singles; and males are less happy than females in both surveys and both years (although the effects are not always statistically significant in the WVS). In 2011, when income is defined in the WVS as relative income, the income variable is replaced by the ELQ variable in the TLSS regression for comparison. In that case, the estimates of the effect of income are found to be similar across the two datasets. Higher relative income is associated with higher happiness in both WVS and TLSS.

The differences are observed in the estimates of age, absolute income and education. First, although the relationship between happiness and age has been found to have a convex shape, the age at which lowest happiness is achieved (using the estimated marginal effects of age and age squared) differs across the two datasets. In the WVS it is in the early 40s, but in the TLSS it is in the early 50s. Secondly, the marginal effect of income on the probability of being very happy is quite different across the two surveys. In the 2007 regressions, absolute income is used. To match the cut-off points of the TLSS brackets as closely as possible, ten income brackets in the WVS are collapsed to six. Even after the adjustment, the estimates of marginal effects are still quite different. In the TLSS, higher income brackets are always associated with higher happiness; however, in the WVS the estimates are the highest for the 2nd and 3rd brackets and negative for the highest two income categories, which is surprising. Thirdly, we observe differences in the estimates of educational attainment across the two surveys even after the educational categories are re-defined to match exactly. Although in 2011 being more educated generally increases the probability of being very happy (no statistical significance in the WVS), in 2007 the results are inconsistent across the two surveys. In the TLSS data those with a post-graduate degree are less likely to be very happy; in the 2007 WVS they are more likely to be very happy.

As a robustness check for the statistical significance of the estimates, the Huber/White heteroskedasticity consistent standard errors are estimated and presented in separate columns ("Robust SE") in Tables 7 and 8. Clearly, the results change very little. Only in a few cases the degree of statistical significance is reduced when robust standard errors are used.

As another robustness check, the estimations are repeated for males and females separately (see Tables 9, 10). The overall impact of the division of the sample into two is the reduction in the sample size and the consequent drop in the degree of statistical significance. Otherwise, the findings are qualitatively very similar for males and females.

Table 9

Comparison of the estimates for males and females and across the two datasets (V

Marginal effects for the probability of being very happy, their robust standard errors and

	World values survey (WVS)				Turkish	life satisfa
	Males		Females		Males	
	ME	SE	ME	SE	ME	SE
Age	-0.021	0.008**	-0.018	0.008**	-0.016	0.003***
Age squared	0.000	0.000***	0.000	0.000*	0.000	0.000***
Income group						
2	0.003	0.056	0.078	0.052	0.093	0.028***
3	0.011	0.064	0.076	0.068	0.152	0.028***
4	0.034	0.065	0.007	0.056	0.159	0.028***
5	-0.023	0.061	-0.093	0.062	0.214	0.028***
6	-0.146	0.068**	0.120	0.121	0.238	0.025***
Marital status						
Married	0.046	0.049	0.183	0.056***	0.088	0.027***
Widow/widower	-0.073	0.251	0.068	0.105	0.001	0.071
Divorced	-0.295	0.040***	0.064	0.095	-0.089	0.068
Separated	-0.321	0.035***	0.188	0.308	-0.120	0.090
Education						
Primary school	0.120	0.076	0.027	0.056	0.016	0.031
High school	0.061	0.082	0.067	0.068	0.031	0.034
University	0.139	0.092	0.018	0.098	0.005	0.039
>University	0.739	0.072***	0.311	0.217	-0.081	0.096
Unemployed	-0.084	0.065	-0.327	0.141**	-0.100	0.030***
Housewife		-0.001	0.069			-0.008
Student	-0.041	0.067	0.004	0.097	-0.011	0.055
Retired	-0.065	0.064	-0.057	0.141	-0.006	0.020
N	670		648		2865	
Log likelihood	-6940 6		-6820 4		-3345 5	

Log nkomiood	U/TU.U	00 2 0.T	JJTJ.J			
Pseudo R2	0.033	0.026	0.900			
Source Author's calculations. See notes to Table 7						

Table 10

Comparison of the estimates for *males* and *females* and across the two datasets (V Marginal effects for the probability of being very happy, their *robust* standard errors and

	World values survey (WVS)				Turkish life satisfa		
	Males	Males		Females		Males	
	ME	SE	ME	SE	ME	SE	
Age	-0.012	0.010	0.001	0.008	-0.015	0.002***	
Age squared	0.000	0.000	0.000	0.000	0.000	0.000***	
Income group		·		·			
1	_		_		-0.082	0.070	
2	0.189	0.192	0.268	0.144*	0.059	0.065	
3	0.090	0.105	0.187	0.116	0.060	0.062	
4	0.112	0.101	0.178	0.111	0.155	0.060***	
5	0.195	0.102*	0.201	0.111*	0.180	0.060***	
6	0.177	0.102*	0.249	0.113**	0.219	0.060***	
7	0.238	0.103**	0.201	0.110*	0.233	0.060***	
8	0.234	0.110**	0.299	0.118**	0.255	0.060***	
9	0.299	0.117**	0.349	0.122***	0.217	0.076***	
10	0.509	0.203**	0.490	0.192**	0.251	0.063***	
Marital status							
Married	0.087	0.047*	0.151	0.049***	0.139	0.022***	
Widow/widower	0.128	0.123	0.014	0.078	0.019	0.050	
Divorced	-0.035	0.140	-0.019	0.079	-0.186	0.059***	
Separated	-0.303	0.035***	-0.237	0.052***	-0.243	0.074***	

Education						
Primary school	0.103	0.081	0.009	0.051	0.100	0.035***
High school	0.095	0.087	-0.026	0.058	0.087	0.038**
University	0.170	0.094*	-0.036	0.076	0.122	0.038***
Unemployed	-0.133	0.069*	-0.107	0.114	-0.085	0.026***
Housewife	_		-0.042	0.051	_	
Student	-0.048	0.077	0.024	0.084	0.008	0.034
Retired	-0.052	0.081	-0.055	0.088	0.047	0.019**
N	758		807		3319	
Log likelihood	-7980.2		-8820.3		-3722.1	
Pseudo R2	0.0649		0.0145		0.9	

Source Author's calculations. See notes to Table 8

One explanation for the observed differences in estimates is the differences in the frequency distribution of the answers given to the happiness question. Respondents are offered different alternative answers to the happiness question in the two surveys. Both the scale and the wording of the question vary across surveys. In one survey the scale is 1−4, in the other it is 1−5. In both surveys the highest happiness category is "Very happy"; however, ≤less than 10 % of the respondents in the TLSS picked that category, whereas in the WVS 37−38 % did. It appears that the scale of the happiness question matters more than the wording of the categories. The wording of the top category is the same, but a much smaller share of the respondents picked that category when five categories were available than when four categories were available.

To attempt to fix this problem, two methods are used. First, the categories are collapsed to match the frequency distribution as well as possible across surveys. This method is applied here. A binary happiness variable (happy versus not happy) is created. The shares in the TLSS are 89 and 11 % in 2007, 90 and 10 % in 2011. In the WVS, they are 86 and 14 % in 2007 and 84

and 16 % in 2011. Then, the happiness regression is estimated by logit. The results are qualitatively the same as before. (Estimates are available upon request).

Secondly, a rescaling is done. DeJonge et al. (2014) suggest rescaling to make different scales of the happiness question consistent. One way of rescaling is to assign numbers that are stretched to a common range such as from 0 to 10, so that 0 is assigned to lowest happiness and 10 to highest happiness, and all intermediate options are given equally distanced numbers in between (called the 'linear stretch'). In the TLSS, the alternatives are numbered 0-2.5-5-7.5-10. In the WVS, they are numbered 0-3.3-6.6-10. With the newly assigned values of the happiness variable, we can estimate the happiness regressions using ordinary least squares. Once again, the results are qualitatively the same as before. (Estimates are available upon request.)

5. Discussion and Conclusion

This paper has focused on the determinants of happiness in Turkey in the recent years. Although a large literature exists on what makes people happy, only a few published studies exist on the economics of happiness in Turkey. The few studies that exist agree on only some of the findings. Therefore, a natural question to ask, as done in this paper, is whether the association of happiness with its potential determinants is stable across time and across samples. For that purpose, the latest waves of two nationally representative micro datasets, the TLSS and the WVS for Turkey (WVS), in years 2007 and 2011, are used and regression analyses are conducted.

After showing that the association of happiness with its potential determinants usually depends on the time period, the set of control variables and the sample used, the paper asks whether the findings would be similar across samples if the time period and the set of variables were equalized. The results are similar across samples only for some factors. The similarities in the two sets of estimates are that the unemployed are less happy than others; married individuals are happier than singles; divorced, separated or widowed individuals are less happy than singles; and males are less happy than females (although the effects are not always statistically significant in the WVS) in both surveys and both years. The findings on income are similar across

surveys when income is defined as relative income and not as absolute income. In both WVS and TLSS, higher relative income is associated with higher happiness.

The differences are observed in the estimates of age, absolute income and education. First, the age at which lowest happiness is achieved is in the early 40s in the WVS, but in the early 50s in the TLSS. Secondly, the marginal effect of absolute income on the probability of being very happy is quite different across the two surveys. Thirdly, the estimates of educational attainment are different across the two surveys. Although in 2011 being more educated generally increases the probability of being very happy (no statistical significance in the WVS), in 2007 the results are inconsistent across the two surveys. In the TLSS data those with a post-graduate degree are less likely to be very happy; in the 2007 WVS they are more likely to be very happy.

It has been shown in the literature that controlling for individual unobserved effects can change the findings about the determinants of happiness (Ferreri-Carbonell and Frijters 2004). Future efforts to uncover what makes us happy should focus on collecting panel data to allow for controlling for such effects. Furthermore, with panel data, it would be possible to ask whether changes in some factors, such as family, work, or social environment, yield any changes in happiness. Currently, no panel data exist that include questions on happiness or life satisfaction in Turkey. In case the collection of panel data is infeasible, the least that can be done is to revise the surveys to incorporate questions on individual unobserved effects such as personality. Psychology literature suggests that personality traits can be learned aboutrevealed by asking a small number of questions to the respondents [see, for example, Gosling et al. (2003)]. The benefits from including a few questions on personality may very well exceed the costs.

One result that emerges from this paper is the importance of being consistent in the scale and the wording of the happiness questions across surveys. In our case, the TLSS has five alternative answers and the WVS has four. Although the top alternative is 'very happy' in both surveys, the stark difference in the shares of respondents who have chosen that alternative is obvious. Different scales adopted by the two surveys generate two different frequency

distributions of the answers to the happiness question, which might be at least partly responsible from the inconsistency of the estimates across surveys. Although the definitions of control variables can be unified with simple fixes, rescaling the dependent variable is not easy if possible. The recommendation is to unify both the scale and the wording of the happiness and life satisfaction questions across surveys.

It is also implicitly assumed in the literature that the factors related to the wider economic, social and political environment we live in are the same for all individuals in a particular year; however, the assumption may be wrong. For instance, economic and political shocks, such as crises, elections, and migration, may have different effects across provinces. The provincial variation in the impact of shocks could be exploited to measure their effect on happiness, if such data existed. The collection of data representative at the province level would require sample sizes to considerably expand. Therefore, the costs should be evaluated against potential benefits.

Another challenge for future research is to identify the factors that 'cause' happiness, which requires going beyond reporting of correlations and identifying the direction of causality with proper techniques. It is possible that reverse causality exists between happiness and factors such as health, education, and allocation of time to leisure. The investigation of causality would require an exogenous change in these variables, which happens only from time to time. An example is an education reform that changes educational attainment suddenly and unexpectedly.

The conclusion is that the results of the paper are useful in understanding the correlates of happiness and life satisfaction. It is evident, however, that only some of the findings are robust to changes in time and the sample. Several recommendations emerge from the analyses conducted in the paper. First, data collectors could be more sensitive to being consistent in the scale and the wording of the happiness questions across surveys. Secondly, the data could have a panel dimension. Thirdly, the sample size could be increased to achieve representativeness in regions and provinces. Turkey is a country with a large population and ample diversity across groups of people and regions. The variation across geographical areas could be employed to better understand the determinants of happiness and life satisfaction. It is inevitable

that a renovation of the data collection process will increase costs, yet the increase in costs should be weighed against the substantial increase in benefits.

Acknowledgments

The author gratefully acknowledges financial support provided by Turkish Scientific and Technological Council (TUBITAK) (Project Number 111K501). Comments and suggestions received from Aysıt Tansel are highly appreciated. Thanks to Ekrem Ozan Bahçeci and Nejat Gökhan Okatan for providing research assistance. The usual disclaimer applies.

References

Blanchflower, D. G., & Oswald, A. J. (2004). Well-being over time in Britain and the USA. *Journal of Public Economics*, 88, 1359–1386.

Caner, A. (2014). Happiness, comparison effects, and expectations in Turkey. *Journal of Happiness Studies*,. doi:10.1007/s10902-014-9562-z.

Caporale, G., Georgellis, Y., Tsitsianis, N., & Yin, Y. (2009). Income and happiness across Europe: Do reference values matter? *Journal of Economic Psychology*, 30, 42–51.

Clark, A., Frijters, P., & Shields, M. (2008). Relative income, happiness and utility: An explanation for the Easterlin paradox and other puzzles. *Journal of Economic Literature*, 46, 95–144.

Clark, A. E., & Oswald, A. J. (1998). Comparison-concave utility and following behaviour in social and economic settings. *Journal of Public Economics*, 70, 133–155.

Davis, E. E., & Fine-Davis, M. (1991). Social indicators of living conditions in Ireland with European comparisons. *Social Indicators Research*, *25*, 103–364.

DeJonge, T., Veenhoven, R., & Arends, L. (2014). Homogenizing

responses to different survey questions on the same topic: Proposal of a scale homogenization method using a reference distribution. *Social Indicators Research*, 117(1), 275–300. doi:10.1007/s11205-013-0335-6.

Dolan, P., Peasgood, T., & White, M. (2008). Do we really know what makes us happy? A review of the economic literature on the factors associated with subjective well-being. *Journal of Economic Psychology*, 29, 94–122.

Dumludag, D. (2013). Life satisfaction and income comparison effects in Turkey. *Social Indicators Research*, 114(3), 1199–1210.

Easterlin, R. (1974). Does economic growth improve the human lot? Some empirical evidence. In P. A. David & M. W. Reder (Eds.), *Nations and households in economic growths* (pp. 89–125). New York: Academic Press.

Ekici, T., & Koydemir, S. (2013). Social capital, government and democracy satisfaction, and happiness in Turkey: A comparison of surveys in 1999 and 2008. *Social Indicators Research*, 118, 1031–1053.

Fahey, T., & Smyth, E. (2004). Do subjective indicators measure welfare? Evidence from 33 European societies. *European Societies*, 6(1), 5–27.

Ferrer-i-Carbonell, A. (2005). Income and well-being: An empirical analysis of the comparison income effect. *Journal of Public Economics*, 89, 997–1019.

Ferrer-i-Carbonell, A., & Frijters, P. (2004). How important is methodology for the estimates of the determinants of happiness. *The Economic Journal*, 114, 641–659.

Flouri, E. (2004). Subjective well-being in midlife: The role of involvement of and closeness to parents in childhood. *Journal of Happiness Studies*, 5, 335–358.

Frey, B. S., & Stutzer, A. (2002). What can economists learn from

happiness research? Journal of Economic Literature, 40(2), 402-435.

Frijters, P., Liu, A. Y., & Meng, X. (2012). Are optimistic expectations keeping the Chinese happy? *Journal of Economic Behavior & Organization*, 81, 159–171.

Gao, W., & Smyth, R. (2011). What keeps China's migrant workers going? Expectations and happiness among China's floating population. *Journal of the Asia Pacific Economy*, 16(2), 163–182.

Gitmez, A. S., & Morcol, G. (1994). Socio-economic status and life satisfaction in Turkey. *Social Indicators Research*, *31*(1), 77–98.

Gosling, S. D., Rentfrow, P. J., & Swann, W. B, Jr. (2003). A very brief measure of the Big-Five personality domains. *Journal of Research in Personality*, 37, 504–528.

Graham, C. (2010). Happiness around the World: The paradox of happy peasants and miserable millionaires. Oxford: Oxford University Press.

Graham, C., & Pettinato, S. (2002). Frustrated achievers: winners, losers, and subjective well being in the new market economies. *Journal of Development Studies*, 38, 100–140.

Guriev, S., & Zhuravskaya, E. (2009). (Un)happiness in transition. *Journal of Economic Perspectives*, 23(2), 143–168.

Kahneman, D., & Krueger, A. (2006). Developments in the measurement of subjective wellbeing. *Journal of Economic Perspectives*, 20(1), 3–24.

Kahneman, D., Wakker, P. P., & Sarin, R. (1997). Back to bentham? Explorations of experienced utility. *The Quarterly Journal of Economics*, 112(2), 375–405.

Knight, J., Song, L., & Gunatilaka, R. (2009). Subjective well-being and its determinants in rural China. *China Economic Review*, 20(4), 635–649.

Layard, R. (2005). *Happiness: Lessons from a new science*. London: Penguin Books.

Layard, R., Mayraz, G., & Nickell, S. (2010). Does relative income matter? Are the critics right? In E. Diener, J. F. Helliwell, & D. Kahneman (Eds.), *International differences in well-being*. Oxford: Oxford University Press.

Ng, Y.-K. (1996). Happiness surveys: Some comparability issues and an exploratory survey based on just perceivable increments. *Social Indicators Research*, *38*, 1–27.

OECD. (2011). How's life?: Measuring well-being. OECD Publishing. doi: 10.1787/9789264121164-en.

Selim, S. (2008). Life Satisfaction and Happiness in Turkey. *Social Indicators Research*, 88, 531–562.

Sen, A. K. (1986). The Standard of Living. S. McMurrin içinde, Tanner Lectures on Human Values, Vol. VII. Cambridge University Press, Cambridge

Senik, C. (2004). Income distribution and well-being: What can we learn from subjective data? *Journal of Economic Surveys*, 19, 43–63.

Senik, C. (2009). Direct evidence on income comparisons and their welfare effects. *Journal of Economic Behavior & Organization*, 72, 408–424.

Stark, O., & Taylor, J. E. (1991). Migration incentives, migration types: The role of relative deprivation. *The Economic Journal*, 101(408), 1163–1178.

Stutzer, A. (2004). The role of income aspirations in individual happiness. *Journal of Economic Behavior & Organization*, *54*, 89–109.

Van Praag, B., & Ferrer-i Carbonell, A. (2008). Happiness quantified: A

satisfaction calculus approach. USA: Oxford University Press.

Veenhoven, R. (2000). The four qualities of life, ordering concepts and measures of the good life. *Journal of Happiness Studies*, 1, 1–39.

Veenhoven, R. (2006). New directions in the study of happiness: United States and international perspectives. Conference paper. University of Notre Dame, USA.

Veenhoven, R. (2013). Happiness in Turkey (TR). Retrieved July 1, 2013, from world database of happiness, Erasmus University Rotterdam, The Netherlands: http://worlddatabaseofhappiness.eur.nl

Zapf, W., Glatzer, W., Noll, H. H., Habich, R., Berger-Schmitt, R., Bruer, S., et al. (1987). German social report: Living conditions and subjective well-being, 1978–1984. *Social Indicators Research*, 19, 1–171.

The following table shows the p-values for the test of equality of income bracket coefficient estimates in the regression in Table 3, column (1). For example, the hypothesis beta[income 2] = beta[income 3] has a p value of 0.062.

Brackets	2	3	4	5	6
2		0.062	0.017	0.000	0.000
3			0.699	0.000	0.000
4				0.000	0.000
5					0.321

¹ In fact, the question in the TLSS assigns 1 to "Very happy" and 5 to "Very unhappy". The alternatives have been re-numbered so that higher values correspond to greater happiness. This facilitates the interpretation of the estimates.

² The data files used in the study are "WV5_Data_Turkey_2007_stata_v_2014_04_28.zip" and "WV6_Data_Turkey_2011_stata_v_2014_04_28.zip".

³ The alternatives have been re-numbered so that higher values correspond to greater happiness.

⁴ In 2007, weighted pairwise correlation is 0.589 and Spearman's rank order correlation 0.568. In 2011, the corresponding figures are 0.537 and 0.5103. The p-values for tests of independence are almost zero.

⁶ Rescaling will not change the estimates in the ordered logit model, since the model is not sensitive to the cardinality of the alternatives and only the order matters.